Author’s response to reviews

Title: Primary vaginal calculus in a woman with urogenital sinus anomaly: A case report

Authors:
Qiong Xu (5512040@zju.edu.cn)
yu zou (zouyuzju@zju.edu.cn)

Version: 4 Date: 21 Aug 2020

Author’s response to reviews:

Editor, BMC urology

Dear Editor

We are truly grateful to the editorial comments and suggestion, and the itemized response to the editorial comments is attached. Many thanks for your suggestion. We have made careful modification on the original manuscript. We hope the new manuscript will meet your magazine’s standard.

Thanks very much again for your attention to our paper again, thank you for your help to our paper processing.

Editor Comments:

1. Overlap:
We note that the current submission contains some textual overlap with other previously published works, in particular:
This overlap exists throughout your manuscript.
While we understand that you may wish to express some of the same ideas contained in this publication, please be aware that we cannot condone the use of text from previously published work.
Please rephrase these sections to minimise overlap and if there is overlap in the Methods section, please ensure that you summarize the methods and cite the original source.
Please be informed that we cannot proceed with handling your manuscript before this issue is resolved.
Response: We have revised the entire manuscript, including the abstract, case present, conclusion and references.
2. Figures:
We note that Figures 1a, 1b, 1c, Figures 2a, 2b, and 2c and Figures 3a and 3b are not cited in the text of your manuscript. Please ensure that all figures, tables and supplementary/additional files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.
Response: Figures 1a, 1b, 1c, Figures 2a, 2b, and 2c and Figures 3a and 3b have been cited in the text of the manuscript. (See line 13-14, line 16, line19, line 23 and line 25 from top, Case presentation). In addition, a, b and c had been added in Figure Legend.

3. Copy edit:
We recommend that you ask a native English speaking colleague to help you copy edit the paper. If this is not possible, you may need to use a professional language editing service. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication.
Response: We have revised the entire manuscript.

4. Remove CARE checklist:
Please remove the CARE checklist as this is no longer required at this stage of the editorial process.
Response: The CARE checklist has been removed.

5. Clean Copy:
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Response: The uploaded manuscript is the final, clean version.

Thank you for your attention in processing the manuscript and we are looking forward to your favorable decision. Correspondence should be addressed to Yu Zou at the following address and email address.

Radiology department, Women’s Hospital, Zhejiang University School of Medicine
No. 1 Xueshi Road
Hangzhou 310006, China
Tel: +86 13777556095
Email: zouyuzju@zju.edu.cn

Sincerely yours

Yu Zou